

September 8, 2021

Dear Prof. Lian-Sheng Ma and reviewers:

Thank you for your letter and comments concerning our manuscript entitle “Increased risk of colorectal neoplasia in inflammatory bowel disease patients with post-inflammatory polyps: A systematic review and meta-analysis” (ID: 70554). We are very pleased that our paper has received your preliminary acceptance. We have further revised the content and format of the manuscript according to the manuscript revision guidelines and requirements. We have sent the revised manuscript to a professional English language editing company to polish the manuscript further and the language certificate has been submitted. We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper. We appreciate for Editors/Reviewers’ warm work earnestly, and sincerely hope this manuscript will be finally acceptable to be published on World Journal of Gastrointestinal Oncology. Thank you very much for all your help and looking forward to hearing from you soon.

Yours Sincerely,

Xiaosheng He

The 6th Affiliated Hospital, Sun Yat-Sen University,  
No 26, YuanCun ErHeng Road, TianHe District,  
510655, Guangzhou, China.

E-mail: hexsheng@mail.sysu.edu.cn